

Citation: Ullah H, Nagelkerken I, Goldenberg SU, Fordham DA (2018) Climate change could drive marine food web collapse through altered trophic flows and cyanobacterial proliferation. PLoS Biol 16(1): e2003446. <u>https://doi.org/10.1371/journal. pbio.2003446</u>

Academic Editor: Michel Loreau, National Centre for Scientific Research (CNRS), France

Received: June 28, 2017

Accepted: December 5, 2017

Published: January 9, 2018

Copyright: © 2018 Ullah et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: Environment Institute https://www. adelaide.edu.au/environment/. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. Australian Research Council http:// www.arc.gov.au/ (grant number FT120100183). The funder had no role in study design, data collection and analysis, decision to publish, or RESEARCH ARTICLE

Climate change could drive marine food web collapse through altered trophic flows and cyanobacterial proliferation

Hadayet Ullah¹, Ivan Nagelkerken^{1,2}*, Silvan U. Goldenberg¹, Damien A. Fordham²

1 Southern Seas Ecology Laboratories, School of Biological Sciences, University of Adelaide, Adelaide, Australia, 2 The Environment Institute, School of Biological Sciences, The University of Adelaide, Adelaide, Australia

* ivan.nagelkerken@adelaide.edu.au

Abstract

Global warming and ocean acidification are forecast to exert significant impacts on marine ecosystems worldwide. However, most of these projections are based on ecological proxies or experiments on single species or simplified food webs. How energy fluxes are likely to change in marine food webs in response to future climates remains unclear, hampering forecasts of ecosystem functioning. Using a sophisticated mesocosm experiment, we model energy flows through a species-rich multilevel food web, with live habitats, natural abiotic variability, and the potential for intra- and intergenerational adaptation. We show experimentally that the combined stress of acidification and warming reduced energy flows from the first trophic level (primary producers and detritus) to the second (herbivores), and from the second to the third trophic level (carnivores). Warming in isolation also reduced the energy flow from herbivores to carnivores, the efficiency of energy transfer from primary producers and detritus to herbivores and detritivores, and the living biomass of detritivores, herbivores, and carnivores. Whilst warming and acidification jointly boosted primary producer biomass through an expansion of cyanobacteria, this biomass was converted to detritus rather than to biomass at higher trophic levels-i.e., production was constrained to the base of the food web. In contrast, ocean acidification affected the food web positively by enhancing trophic flow from detritus and primary producers to herbivores, and by increasing the biomass of carnivores. Our results show how future climate change can potentially weaken marine food webs through reduced energy flow to higher trophic levels and a shift towards a more detritusbased system, leading to food web simplification and altered producer-consumer dynamics, both of which have important implications for the structuring of benthic communities.

Author summary

Healthy marine ecosystems are crucial for people's livelihoods and food production. Global climate stressors, such as warming and ocean acidification, can drastically impact the structure and function of marine food webs, diminishing the production of goods and



preparation of the manuscript. Australian Research Council <u>http://www.arc.gov.au/</u> (grant number FT140101192 (to support D.A.F)). D.A.F was supported by Future Fellowship grant # FT140101192. The funder had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: MTE, metabolic theory of ecology; *OA*, elevated CO₂; *OAT*, elevated CO₂ and temperature; *T*, elevated temperature.

services. Our ability to predict how future food webs will respond to a changing environment is limited by our understanding of species responses to climate change, which are often tested in isolation or in simplified experimental designs. More realistic predictions of the impacts of climate change on ecosystems requires consideration of entire species communities, including the species interactions that can buffer or exacerbate these impacts. We experimentally tested the effects of warming and acidification, both individually and in combination, on a benthic marine food web in a near-natural ecological setting. Energy flow from the first trophic level (primary producers and detritus) to the second (herbivores), and from the second to the third trophic level (carnivores) was quantified under these different regimes. We show that warming, either alone or in combination with acidification, can constrain productivity to the bottom of the food web by enhancing cyanobacterial biomass and reducing energy flow to higher trophic levels, thus lowering energy transfer efficiency between producers and consumers. In contrast, increased ocean acidification alone showed a positive effect on herbivores and carnivores. Our finding is important because it demonstrates that future warming could drive marine food web collapses to potentially simplified and less productive coastal systems.

Introduction

Forecasting the effect of global change on ecosystem functioning is a major challenge in ecology [1], partly because future climate shifts are likely to reorganize complex food webs, generating novel communities composed of new combinations of species [2]. In marine ecosystems, multiple anthropogenic stressors are already eroding biodiversity by changing the composition of species [3] and affecting rates of biomass transfer through ecological networks, resulting in altered food web organisation and dynamics [4,5]. While overexploitation is largely responsible for altering the structure and functioning of many ecosystems [6], global warming is forecast to amplify these effects, having serious consequences for the health and sustainability of marine ecosystems [7]. Despite many studies showing a potential detrimental effect of climate change on biodiversity [2], we still lack a strong and coherent theoretical and empirical foundation for understanding how species communities are likely to respond to global change [8].

Marine ecosystem functioning is maintained by the flow of energy from primary producers at the base of food webs through intermediate consumers to top predators, as well as via cycling of materials such as nutrients within the ecosystem [7,9]. Disturbances such as habitat modification can decouple, alter, or concentrate energy flows towards a smaller number of species and erode resilience by removing alternative feeding pathways in the food web [10]. This can drive trophic food webs to shift states and potentially collapse [11]. In this context, climate change can independently affect, or synergistically amplify, the effect of other disturbances such as habitat degradation, species overexploitation, and species invasions [12]. This can reconfigure future food webs through major structural changes, like shifts in the number of trophic groups and links that connect species at the top of the food chain to basal species, which can result in altered flows of energy and shifts in the biomass of key functional groups, leading to biodiversity loss, with potentially serious implications for ecosystem functioning [13].

Global warming and ocean acidification are already affecting the physiology, behaviour, phenology, demography, abundance, and distribution of many marine species [14]. Elevated temperature (T) affects fish performance and growth through increasing metabolic rates and respiratory demands, leading to a reduced aerobic scope for important life-supporting

activities such as feeding, somatic growth, maturation, and predator avoidance [15–17]. Ocean acidification raises the energetic costs involved with calcification and acid-base regulation [18,19] and can impair neural functioning [20]. Besides such direct effects, there is a suite of indirect effects that can impact species persistence and diversity under global change [21].

The survival of a species or a group of species in an ecosystem depends on how well they respond to dynamic changes in productivity, in terms of direction (i.e., positive or negative) and strength. However, the responses of species to global change are not individual-based; they are connected through biotic interactions within and across trophic levels [22,23]. Importantly, the energy flow (biomass fluxes) to higher trophic levels is determined by various biological interactions (e.g., predator-prey relationships, competition, facilitation, and mutualism) of species that are directly or indirectly linked to proximate trophic levels [7,24]. For example, climate change can weaken the energy transfer efficiency between primary producers and consumers by reducing feeding performance with potential impacts further up the food chain [25]. Although ocean warming and acidification can boost basal productivity [26,27], it does not necessarily result in an increase in secondary productivity [28]. The propagation of production through trophic levels could be modified by food chain length [29] or with an increase in the dominance of herbivory-resistant primary producers [30,31]. Cyanobacteria, in the form of toxin-producing deleterious phytoplankton, can potentially divert productivity into alternate food web pathways, which are unavailable to higher trophic levels. Cyanobacteria have been forecast to increase in dominance as a result of global warming [29]. Conversely, in line with metabolic theory [32], increased food demand in predators can intensify topdown control of their prey populations [16], influencing body size-driven metabolic differences in food web structure [33]. Thus, a dichotomy between enhanced bottom up control through an increase in herbivory-resistant primary producers and enhanced top-down control due to higher metabolism-driven energetic demand in predators can jeopardise intermediate trophic levels and weaken food web stability. Such a mismatch can alter trophic energy flow dynamics [25,34], with the consequences expected to be greater in food webs with three trophic levels or more [29].

Forecasting the effects of climate change at the ecosystem level requires holistic approaches that consider complex ecological communities with multiple functional groups or species of different trophic levels [35]. Large-scale mesocosm experiments are ideal for such approaches and have the potential to enhance our understanding of the ecological consequences of climate change on the sudden expansion of opportunistic species, species extinction risk, community structure, and ecosystem functions [36].

We did a community-level manipulation of a temperate marine food web, consisting of 17 functional groups (ranging from primary producers to herbivores to carnivores across 3 broad trophic levels). We maintained these groups in an indoor mesocosm experiment divided into 4 treatments: elevated CO_2 (*OA*), elevated temperature (*T*), elevated CO_2 and temperature (*OAT*), and ambient controls (present-day levels of pCO_2 and temperature), each with 3 replicate mesocosms per treatment. We achieved an elevated pCO_2 of approximately 900 ppm (pH = 7.89) and warming of +2.8 °C, which represent the conditions predicted for the end of this century, following a business-as-usual emission scenario (RCP8.5) [37]. We used an ecosystem modelling tool (Ecopath) widely used to characterise quantitative food web structures and pathways of energy flows in aquatic ecosystems [38]. The Ecopath model is built on a system of linear equations and creates a static mass-balanced snapshot of the resources in a given ecosystem according to biomass estimates and food consumption relationships of functional groups that represent the organisms in a food web. The quantitative description of food web properties is essential to advance our understanding of ecosystem structure and functioning at a community level [39]. Using the Ecopath approach, we tested whether: (1) global warming

and ocean acidification enhance energy fluxes through bottom-up effects that stimulate primary productivity, or (2) global warming and ocean acidification allow opportunistc groups to proliferate and divert productivity into alternative pathways, as well as (3) biomass of lower trophic levels and detritus will dominate the future structure of marine food webs due to reduced energy transfer efficiencies to higher trophic levels. We also test whether synergies between ocean acidification and increasing temperature are likely to amplify the effect of global warming on marine ecosystems. Thus, through a combination of experimental and modelling approaches, we provide new evidence for altered energy flow (biomass fluxes) and transfer efficiency through food webs due to global change stressors, which is crucial for understanding the potential effects of climate change on marine food web structure and functioning.

Results

Neither warming nor acidification affected the energy flow originating from primary producers and detritus at trophic level 1 (Fig 1; S1 Table). The combined effect of warming and acidification (*OAT*) (p = 0.003; post hoc energy flow: *OAT* < control) reduced the energy flow from trophic level 1 to trophic level 2. In contrast, energy flow was higher in the *OA*-only treatment compared to the controls (p = 0.011; post hoc energy flow: high CO₂ > control). Warming (*T* and *OAT*) also reduced the energy flows from trophic level 2 to trophic level 3 (ANOVA, $F_{1, 8} = 43.06, p < 0.001$).

The individual functional groups at trophic level 2 and trophic level 3 showed variable responses to warming and acidification. For example, functional groups such as meiobenthos, copepods, small epifaunal invertebrates, and filter feeders experienced significantly lower energy flow from trophic level 1 to trophic level 2 under warming (*T*), while macroinvertebrates experienced reduced flow under the combination of warming and acidification (*OAT*) (S1 Fig; S4 Table). Furthermore, warming significantly reduced the capacity of transferring energy flows of omnivorous, filter feeding, and benthic carnivorous fish, while benthic carnivorous and carnivorous fish experienced an increase in flow under acidification from trophic level 2 to trophic level 3 (S2 Fig, S5 Table).

The reduced energy flow from trophic levels 1 to 2 under the combined warming-acidification treatment (*OAT*) coincided with a negative effect of warming (*T* and *OAT*) on energy transfer efficiency between levels 1 and 2 (ANOVA, $F_{1,8} = 11.22$, p = 0.010) (Fig 1; S1 Table). In contrast, *OA* had no effect on energy transfer efficiency between these levels. Energy transfer efficiency between trophic levels 2 to 3 was not affected by either warming or acidification.

Whilst the combined effect of warming and acidification enhanced the biomass of primary producers (p = 0.001; post hoc living biomass: OAT > control) and acidification enhanced secondary consumer biomass (p = 0.034; post hoc living biomass: OA > control), warming (T and OAT), irrespective of ocean acidification, caused a decrease in living biomass at trophic levels 2 and 3 (Fig 2; S2 Table). Warming (T and OAT) induced higher cyanobacterial biomass (ANOVA, $F_{1, 8} = 19.90$, p = 0.002), replacing palatable turf algae at trophic level 1 (Fig 3a, S3a Table). Consequently, energy was not transferred to successive trophic levels through consumption but accumulated as detrital biomass (ANOVA, $F_{1, 8} = 9.12$, p = 0.017) (Fig 3b; S3b Table) at the bottom of the food web. The system became less efficient in recycling the accumulated detrital biomass under warming (S3 Fig; S3c Table; ANOVA, $F_{1, 8} = 9.31$, p = 0.016), suggesting a collapse at the base of the food web.

Discussion

Our study provides strong empirical evidence that global warming has the capacity to drive a collapse in marine food webs by altering energy flows between successive trophic levels. In our



Fig 1. The effects of future climate on absolute flows and transfer efficiency between successive trophic levels of the mesocosm food web. Absolute flows, shown as line diagrams, refer to the total amount of energy that flows to higher trophic levels through consumption (log10 g Wet Weight/m²/month) aggregated by trophic level. The first trophic level shows flows originating from both the primary producers and detritus, which are transferred to the second and third trophic level through consumption by herbivores and carnivores, respectively. Transfer efficiency, presented as bar charts, refer to the ratio at which absolute flows are transferred from one trophic level to the next. Mean \pm SE are based on n = 3 mesocosms. Significant effects (p < 0.05) within trophic levels are based on two-way ANOVAs (df = 1,8) and are indicated with asterisks. See S1 Table for statistical test outcomes. Means with different lower-case letters indicate significant differences among treatments based on posthoc tests corrected for false discovery rate and done separately for different trophic levels. The distribution of organisms within the mesocosm is reflected as their vertical position inside the graph (ranging from the bottom of the mesocosm to the surface of the water column). Species cliparts were obtained or modified from Openclipart (https://openclipart.org/). C, control; not sig, not significantly different; OA, elevated CO₂; OAT, elevated CO₂ and temperature; T, elevated temperature.

https://doi.org/10.1371/journal.pbio.2003446.g001

ecologically complex benthic mesocosm experiment, the combination of warming and acidification enhanced the biomass of primary producers, but reduced energy flow to herbivores, while warming (irrespective of acidification) reduced energy flow to carnivores at higher trophic levels. Warming also decreased the trophic transfer efficiency between primary producers and herbivores, consequently reducing standing biomass of herbivores and carnivores. Other studies based on the metabolic theory of ecology (MTE) [32], however, suggest that temperature-driven increased primary production is likely to propagate through food webs via strong top-down control [40,41], resulting in greater levels of heterotrophic biomass, relative to autotrophic biomass [4]. In our case, the combination of warming and acidification decoupled increased basal productivity from herbivore production, while warming in isolation reduced predator production, making most of the primary production unavailable further up the food chain. Thus, energy from enhanced primary producer biomass under future climate conditions may not always transfer through to successive trophic levels, but instead can decouple food demand and supply. Such a decoupling may alter dietary preferences of consumers, modifying consumer–prey relationships within food webs.

Our current inability to capture more realistic features of food web responses to global climate change is mostly due to a reliance on short-term, small-scale experiments harbouring single species and lower trophic levels, which provide an ambiguous approximation of





Fig 2. Living biomass of primary producers (trophic level 1), primary consumers (level 2), and secondary consumers (level 3) across functional groups within the mesocosms. The biomass of functional groups with intermediate trophic levels (e.g., trophic level of filter feeders = 2.4) was assigned to the levels 2 and 3 according to their relative contribution to trophic flow (e.g., 60% to level 2 and 40% to level 3). At the third trophic level, the decrease in biomass under T and OAT is primarily driven by filter feeders, while a negative effect was not apparent in most other functional groups such as the fishes (see S2 Fig). Living biomass includes 16 functional groups excluding detritus. Values are means \pm SE across mesocosms (n = 3). Significant interactions or main effects (p < 0.05) within trophic levels are based on two-way ANOVAs (df = 1,8) and are indicated with asterisks. See S2 Table for statistical test outcomes. Means with different lowercase letters indicate significant difference among treatments based on posthoc tests corrected for false discovery rate and done separately for different trophic levels. C, control; OA, elevated CO₂, OAT, elevated CO₂ and temperature; T, elevated temperature.

https://doi.org/10.1371/journal.pbio.2003446.g002

naturally complex food webs [4]. Moreover, few studies of ecological responses to climate change include predator-prey dynamics and competitive interactions or allow for the potential proliferation of opportunistic groups of species, all of which can greatly influence or reverse many of the predicted responses of species and communities to climate change [26]. In contrast, our experimental results account for complex multispecies interactions and biotic processes, providing an improved representation of natural systems and how these are likely to respond to global warming [42,43]. Our experimental data provide insights into how anthropogenic climate change can potentially affect food web dynamics for relatively short-lived taxa. This is because large scale mesocosm experiments such as ours bridge the gap between simplified experimental conditions and the real world [44], providing important opportunities to better understand the likely mechanisms by which primary productivity under certain future climate conditions propagates through the food web.

Increased standing biomass in our benthic mesocosm experiment was most evident for cyanobacteria under warming (irrespective of acidification). Warming is known to enhance the primary productivity of some taxa, particularly of weedy species such as turf algae [45]. Cyanobacteria form a major component of turf algae [46,47] and are predicted to proliferate and expand under eutrophication and climate change [48]. The potential for cyanobacterial



Fig 3. Relative proportion of cyanobacteria (as a percentage) to turf algae in mat-forming algae measured as benthic cover (A). Flows of production (%) to detritus pool relative to primary productivity (B). Mean \pm SE values per mesocosm are given (n = 3). Significant main effects (p < 0.05) are based on two-way ANOVAs (df = 1,8) and are indicated with asterisks. Means with different lowercase letters indicate significant difference among treatments. See <u>S3</u> <u>Table</u> for statistical test outcomes. C, control; OA, elevated CO₂, OAT, elevated CO₂ and temperature; T, elevated temperature.

https://doi.org/10.1371/journal.pbio.2003446.g003

dominance in turf algal assemblages under future warming and acidification has been demonstrated previously using experimental approaches [49]. The effect of warming sea temperatures is expected to be greater in areas that are shallow and nutrient rich [50], providing ideal environmental conditions for cyanobacteria to invade other benthic primary producer groups [51,52]. Some cyanobacteria are known to be toxic and cause localized anoxia and mortality in marine organisms [48]. Cyanobacteria also produce potent allelochemicals that deter feeding and are difficult to control by grazers [53]. Herbivores like macroinvertebrates and small epifaunal invertebrates predominantly feed on mat-forming turf algae rather than cyanobacteria.

Metabolic theory suggests that consumers will show a greater response to temperature than producers [4,54,55]. Therefore, we hypothesised a priori that warming would drive an increase in metabolic and consumption rates for macroinvertebrates in our experiment. However, reduced food availability, brought about by palatable types of turf algae being replaced by unpalatable cyanobacteria, caused food limitation, preventing increased metabolic rates for macroinvertebrates at higher temperatures, suppressing the flow of energy to the second trophic level [16,56]. Furthermore, biomass of major prey groups such as copepods, small epifaunal invertebrates, and filter feeders, which largely form the diet of consumers at the third trophic level, collapsed under the warming treatments, resulting in significantly less energy available for the third trophic level. One of the reasons for this collapse is an excessive predation pressure on primary consumers by species at the third trophic level (i.e., predators) due to

their higher energetic demand [28], which was not matched by any increase in productivity rates of primary consumers under warming. Alternatively, under warming, the consumer-producer relationship at the base of the food web could be nonsynchronous if consumption rates of herbivores peak earlier than the growth rates of producers, creating a mismatch between production and consumption [45]. This means that under certain conditions, even in the absence of herbivory-resistant primary producers, warming-induced metabolic stress of organisms can effectively decouple the consumer-producer relationship if consumer metabolism and consumption cannot keep pace with increasing production.

In our experiment, the collapse of biomass of major functional groups under warming played a major part in reducing the transfer efficiency of energy between trophic levels in a food web. A lower transfer efficiency of energy was evident between trophic levels 1 and 2. This is because transfer efficiency depends on both the availability of food, the biomass of all consumers, and their consumption rates. The lower standing biomass at trophic level 2 and reduced palatable food abundance at trophic level 1 collectively brought about lower overall transfer efficiency between these trophic levels under warming. Metabolic theory states that the structure and dynamics of ecological communities are based on the individual metabolism of organisms where the individual metabolic rate is primarily controlled by body size, body temperature, and resource availability [57]. Metabolic theory does not specifically consider the standing biomass of consumers, which is an important component of our model, allowing us to calculate transfer efficiency. Therefore, food web properties, such as transfer efficiency, are difficult to interpret from the perspective of metabolic theory, yet they can have important community-level effects [58].

The replacement of turf algae by cyanobacteria further resulted in a more detritus-based food web under warming. Detritus can be very important for sustaining food webs and ecosystem stability [59], but only when proper recycling occurs within the system [60]. Decreasing detritus recycling in ecosystems correlates with decreased system resilience to perturbations, with lower rates of recycling resulting in slower recovery [60,61]. We show (based on the Finn's cycling index) that warming significantly reduced the detritus recycling capacity of the system. The inability of herbivores to consume enhanced primary producer biomass and the simultaneous failure of detritivores to transfer excessive detritus production to the successive trophic levels resulted in an accumulation of detrital biomass at the base of the food chain. Therefore, warming-induced detritus accumulation, as observed in our study, might have far-reaching ecosystem consequences for future oceans, such as the spreading of 'dead zones' through increased microbial activity and consumption of dissolved oxygen in bottom waters [30].

Reduced secondary and tertiary production has been forecast under future acidification [2]. However, this pattern was not detected in our experiment in energy flows, a proxy for production at different trophic levels. We show that acidification can in fact exert positive bottom-up effects on energy flow towards secondary producers (trophic level 2). Ocean acidification could increase secondary productivity in situations where strong indirect positive effects dampen the direct negative effects of elevated CO₂, i.e., through increased habitat and food, as well as reduced predator abundances [62,63]. Only benthic carnivores and carnivorous fish at trophic level 3 experienced increased energy flows under acidification, which supports the finding of increased productivity of a carnivorous fish (*Favongobius lateralis*) in a previous study done in a similar ecological setting [28]. In contrast, omnivorous fish showed a decrease in flow and this group was the major contributor of energy flow to trophic level 3 under control conditions. Taken together, the positive effects on other groups resulted in no overall significant increase in energy flow to trophic level 3 under acidification, there we quantify secondary and tertiary production, using a more complex (and ecologically realistic) food web, which

better captured real-world community structure and important species interactions, and how these are likely to change in response to future global warming and ocean acidification [8,39,64]. This broader food web model also allows us to quantify energy transfer efficiency across multiple trophic levels; an important ecosystem function which can regulate many ecological processes (i.e., trophic structure, food chain length) and mediate ecosystem services (i.e., fisheries production) [65-67]. Our results indicate that the response of future food webs to ocean acidification and temperature are likely to depend on the localized community composition and consumer-resource interactions of the specific ecosystem.

Although we used one of the most complex benthic mesocosm experiments to date, our approach is not devoid of caveats. For example, difficulties in separating the functional roles of turf and cyanobacteria meant that they were modelled as one functional group 'mat-forming algae'. In our model, we did not consider regular bacteria (other than cyanobacteria) as a separate functional group, but rather considered them under detritus. Thus, detritivores are considered to mainly feed on detritus and its associated bacteria. We opted for not using an extra bacterial compartment because bacteria would largely overshadow any other trophic flows of the system [68]. Our study showed a relatively larger biomass flux between trophic levels 1 to 2 compared to between trophic levels 2 and 3 due to the presence of relatively large-bodied primary consumers (such as herbivorous macroinvertebrates: Bulla quoyii), which were too big to handle for the gape-limited predators in our system. The presence of a wider range of higherorder invertebrate predators, as is the case in natural ecosystems, would have reduced this disproportionally high flux between primary producers and primary consumers by stronger topdown control. However, since the focus of our study was to show relative difference among the climate treatments, the results still provide a valuable quantitative insight into the potential future of some benthic marine ecosystem under two co-occurring global climate stressors.

One of the weaknesses of earlier applications of the Ecopath model were assumptions of 'steady-state' or equilibrium conditions, meaning that the model outputs should only be considered for the period across which the model input parameters are deemed valid [69]. Ecopath modelling approaches now no longer assume steady-state conditions but instead the model parameterizations are based on a mass-balance assumption over a chosen arbitrary period. The mass-balance approach in Ecopath filters for mutually incompatible estimates of flow [69]. Moreover, under the Ecopath modelling approach, we assumed that mortality for a prey equals consumption of a predator and that all prey are equal in terms of energetic content. Additional care needs to be taken when inferences are drawn from ecosystem models built for highly dynamic systems due to likely nonlinearities in important food web properties of some functional groups, operating at fine temporal scales. Nevertheless, since we used multiple sampling points through time and averages based on multiple replicates for our model input, especially for taxonomic groups that have the potential to show large oscillations with environmental fluctuations, the model outputs are likely to be indicative of near-future ecological states. Lastly, model outputs were tested using the PREBAL diagnostic and pedigree index (see Supplementary Methods for details on model and data quality), and confirmed a stable model that is ecologically robust.

In summary, our results suggest that warming has the capacity to drive an energetic collapse at the base of marine food webs, and this effect can propagate to higher trophic levels—subsequently leading to a collapse in species biomass of the entire food web. The underlying mechanism for this collapse is the replacement of preferred turf algae by cyanobacterial biomass that drives the system towards food limitation for herbivores, with detrimental effects on their predators, combined with a switch towards a less efficient detritus-driven system. Several studies have reported an apparent increase in the occurrence of cyanobacteria in marine waters globally as a result of increasing temperatures [52], and regionally in temperate [70], tropical [51,71], and subtropical [72,73] areas. Thus, these findings are particularly important in the context of climate change, as mismatches in trophic dynamics can decouple linkages between trophic levels driving ecosystems towards simplified, less productive systems, with cascading effects on ecosystem resilience and functioning.

Materials and methods

Ethics statement

This research was carried out under the approval of the University of Adelaide animal ethics committee (approval S-2012-193A). All the habitats and organisms collections were permitted by the Minister for Transport and Infrastructure and the Government Department of Primary Industry and Regions SA (exemptions: 9902676 and 9902752).

Experimental design

An indoor mesocosm experiment was maintained from February 2015 to July 2015. A total of 12 circular mesocosms, each holding 1,800 L of water were set up inside a large temperature-controlled room to simulate shallow temperate coastal ecosystems typical of the Gulf St. Vincent, South Australia (S4 Fig). All habitats and organisms used in the mesocosms were collected at a depth of 1–5 m within 60 km distance of the mesocosm facility. Each mesocosm comprised of a mosaic of the 3 primary local habitats (with 2 replicate patches per habitat per mesocosm): rocky reef, seagrass, and open sand [74].

Rocky reefs consisted of natural rocks collected in situ and included attached macrophytes dominated by an assemblage of fucoids (Order Fucales; mainly species belonging to the families Fucaceae and Sargassaceae) and benthic invertebrates. Rocks were selected to be as similar as possible in terms of presence and cover of major fucoid species. Seagrass habitat was mimicked by artificial green polypropylene ribbon harbouring epiphytes and planted into fine silica sand at a depth of 6 cm. The seagrass habitat resembled the most abundant local seagrass *Posidonia* spp. [74] and was incubated in situ for 2 weeks to allow for epiphytic colonization. The circular 'rocky reef' patches and 'artificial seagrass' patches were of equal size (0.42 m diameter). The space in between and around these patches was 'open sand' habitat, comprising fine silica sand with a depth range between 6–25 cm. The open sand and sand within the seagrass patches were additionally seeded with 0.025 m³ natural sediment collected in situ between patches of live seagrass and included all infauna and flora.

Fish and invertebrates were introduced into the mesocosm and represented different feeding guilds (see <u>S6 Table</u> for a list of species associated with the mesocosms, their stocking density, and mean sizes). The fishes were selected based on their high local juvenile abundances in shallow coastal waters during summer, while the gastropods came from the rocks used to build the rocky reef patches and were redistributed evenly among all mesocosms. In the flowthrough system, unfiltered seawater from 1.5 km off-shore and 8 m depth continuously supplied nutrients and planktonic propagules to each mesocosm at 2,300 L day⁻¹. A diffuser was used to form a light circular current in the mesocosms to simulate tidal water movement alternating direction in 6-hour intervals (<u>S4 Fig</u>). A lamp was mounted above each mesocosm with a spectrum close to sunlight, which is roughly equivalent to 72.83 \pm 24.78 µmoles/m²/second Photon flux corresponding to a local water depth of 6–7 m (14/10 light-dark cycle, 30-minute dawn and dusk dimming).

We applied a control temperature of 21.0 °C in our mesocosm experiment corresponding to the average summer temperature based on a 5-year dataset of 2 local temperature loggers (5 m depth, 2010–2015, SA Water). OA at each mesocosm was achieved through a header tank preconditioned to elevated pCO_2 levels using pure CO₂ (control system ACQ110 Aquatronica,

Italy). Additionally, each mesocosm was supported by a 60-L bin bubbled heavily with enriched air at 1,000 ppm pCO_2 (PEGAS 4000 MF Gas Mixer, Columbus Instruments, Columbus, Ohio) or ambient air at 400 ppm pCO_2 , to maintain target levels. Submersible titanium heaters inside the 60-L bins were used for the future warming treatments. Temperature and pH were measured daily (Mettler Toledo SevenGo SG2, calibrated daily; <u>S5 Fig</u>), while salinity was measured fortnightly (SR6 refractometer; Vital Sine). The total alkalinity was also measured fortnightly by Gran titration from water samples (888 Titrando, Metrohm, Switzerland). The diurnal variability in pH (<u>S6 Fig</u>) confirms that our mesocosms were autonomous systems that mimicked natural day-night fluctuations. For a description of other seawater properties, see <u>S7 Table</u>.

Species sampling and processing

The fish community, herbivorous macroinvertebrates, small epifaunal invertebrates, filter feeders, and macro-crustaceans were all sampled and counted at the end of the experiment and their biomass measured as wet weight.

Biomass of tanaids, copepods, and meiobenthos was determined using benthic samplers (6.5 cm in diameter and 2 cm depth filled with 1.5 cm of mesocosm sand, with 2 replicate samplers per mesocosm), which were placed at the bottom of the tanks for about a month, allowing colonization of these species. Samples were collected twice from each mesocosm during the experimental period and pooled for each mesocosm prior to processing. Tanaids, meiobenthos, and copepods were extracted from the sand within the benthic samplers via floatation using a Ludox TM colloidal solution with a specific gravity of 1.18. The animals were then collected using a 120 μ m sieve. Microzooplankton biomass was measured following volumetric method [75] by filtering 400 L of water from each mesocosm through a plankton sampler at the end of the experiment.

Phytobenthos biomass was measured using the same benthic samplers as above. Two samplers were placed in each mesocosm for biomass measurements. A micro-spatula was used to carefully scrape the thin phytobenthic layer from the upper surface (approximately 1 mm thick) of the sand. The remaining sand was filtered through a pre-combusted and pre-weighed Whatman GF/C glass fiber filters to determine the detritus biomass. In the laboratory, photosynthetic pigments were extracted from freeze-dried sand samples (0.3–0.6 g) with 10 ml 90% acetone. After 48 hours of darkness at -20° C, the samples were stirred in a vortex, centrifuged at 3,500 rpm for 15 minutes, and extracts were analyzed in a 6,405 UV/Vis, Jenway spectrophotometer and their concentration calculated [76].

Phytoplankton biomass (measured as Chlorophyll *a*) was quantified based on photosynthetic pigment concentration. Four litres of water were filtered from each mesocosm with Whatman GF/C glass fiber filters of 4.7 cm diameter, and ground and extracted [76]. Samples for both phytobenthos and phytoplankton were collected twice during the experimental period, and the average of both was used as the model input.

To estimate the biomass of macrophytes and mat-forming algae, all habitats (rock, seagrass, and open sand) were sampled at the end of the study period. Their wet weight was determined to the nearest 0.1 mg. Mat-forming algae were defined as a mix of turf and cyanobacteria. The relative cover of cyanobacteria in mat-forming algae was estimated using the Coral Point Count with Excel Extensions (CPCe) Software [77]. In addition, community metabolism was measured as gross oxygen production (mgO₂/m³/min¹) once per mesocosm at the end of the study. Oxygen concentration was measured in 1 minute intervals over at least 30 minutes (HQ40d Portable Meter, sensor LDO101, HachTM). A linear regression model of O₂ production rate (where O₂ concentration was modelled as function of time) was fitted and confirmed

a high level of precision in the measurement of O_2 concentrations across the 12 mesocosms (mean \pm SD, $R^2 = 0.94 \pm 0.04$).

Food web model construction

We built mass-balanced food web models. Trophic links were weighted by material fluxes among functional groups using Ecopath [78]. We modelled energy or mass flow over a 4 month time step based on local summer conditions today and in the future. We then converted and expressed the model produced energy flow results from the experimental period to values/month to make it more comparable to other systems.

Our model used the following input parameters: biomass, represented by the value B; production per unit of biomass, represented by the value P/B; consumption per unit of biomass, represented by the value Q/B; diet matrix; and the model-estimated ecotrophic efficiency, represented by the value EE. The latter is a parameter that is derived from the model, describing the fraction of the productivity that is used in the system. Most of these model parameters were calculated using our empirical mesocosm data, including final biomass (end-of-experiment) and diet composition of consumers of various functional groups (see <u>S8</u>, <u>S9</u> and <u>S10</u> Tables). The Q/B ratio for most of the functional groups was calculated using stomach content analysis and in situ feeding trials that incorporated treatment effects. Therefore, it incorporates the direct effects of temperature on metabolism, accounting for estimates of trophic biomass fluxes and efficiencies. In cases where data were not available, data were derived from empirical equations and published information (see <u>S1 Text</u>).

Flow and transfer efficiency were based on the trophic aggregation routine [69] that aggregates the entire system into discrete trophic levels sensu [79]. The discrete trophic levels start with level I, corresponding to the primary producers and the nonliving, detrital compartments. Strict herbivores or detritivores consequently occupy a position of level II. This is followed by higher-order consumers that are allocated to several discrete trophic positions according to the type and amounts of food that reach them along feeding pathways.

Energy flows were calculated for different trophic levels following [80], where—for example —if a group obtains 40% of its food as an herbivore and 60% as a carnivore, the corresponding fractions of the flow through the group are attributed to the herbivore level and the carnivore level, respectively. The relative flows (these are proportions adding up to 1) were converted to absolute amounts and shown as the net amount of energy that flows to higher trophic levels through consumption ($g/m^2/month$). The 'transfer efficiency' is the percentage of trophic flows at trophic level *n* that is converted into flows at level *n* + 1. The transfer efficiency of a given trophic level (trophic level = *n*) not only depends on the available energy at a given trophic level (trophic level = *n*) but also the standing biomass at the next trophic level (trophic level = *n* + 1). The transfer efficiencies between successive discrete trophic levels were calculated as the sum of the flow that is transferred from any given level to the next higher level, plus exports from the original level relative to the throughput (or input) of the given (originating) trophic level [69,78]. The throughput is the sum of all flows (such as consumption, export, respiration, and flows to detritus) in a given trophic level and represents the 'size of the trophic level in terms of flow' [81].

Calculation of transfer efficiency from trophic level 1 to 2 is not possible without having information on gross primary production or respiration [82]. We measured net productivity and respiration in each mesocosm and used them to estimate the transfer efficiency between trophic level 1 and 2 for each mesocosm food web [69,78]. The initial output for both energy flow and transfer efficiency was obtained for discrete trophic levels I to V. However, for the

simplicity of the model output and better visualization, we pooled the data for trophic levels III to Vand showed this as 1 integrated trophic level (i.e., trophic level 3).

We used the Ecopath pedigree routine to quantify the uncertainty associated with the model parameters by recording the origin and quality of the input data and assigning a value of uncertainty or a confidence interval to each input (e.g., biomass, P/B, Q/B, diets), which are then used to calculate the overall model pedigree index. The pedigree index varies between 0 (low precision information) and 1 (high quality, i.e., obtained from modelled system and highly precise), allowing a description of the quality of the model [78]. Our overall individual model pedigree index of 0.71, and a measure of fit of t = 3.819, indicates a very high quality and robust model compared to 393 previously constructed models from habitats from around the world, for which pedigree values ranged between 0.164 and 0.676 [83]. More details on the parameterization and model computation are given in the Supplementary Methods.

Statistical analysis

The effects of warming and ocean acidification on food web properties (response variables: absolute energy flow, transfer efficiency, and standing biomass) were analysed using two-way ANOVAs. Both climate factors were treated as fixed and orthogonal. Before analysis, normality was checked for all response variables using the Shapiro-Wilk test, and homogeneity of variances was assessed using a Levene's test as well as by evaluating plots of residuals against predicted values. Response variables were \log_{10} transformed prior to analysis if they did not conform to a normal distribution. For significant interactions, posthoc multiple comparisons adjusted by false discovery rate were performed [84]. All data analyses were done with the software package R version 3.2.3 [85].

Supporting information

S1 Fig. Absolute flows (gWWm⁻²month⁻¹) produced by the different functional group at trophic level 2. Mean \pm SE values per mesocosm are given (n = 3). Significant interactions or main effects (p < 0.05) within functional groups are based on two-way ANOVAs (df = 1,8) and are indicated with asterisks. Means with different lowercase letters indicate significant difference among treatments based on posthoc tests corrected for false discovery rate and done separately for different functional group. No Sig = no significance. See <u>S4 Table</u> for statistical test outcomes.

(TIF)

S2 Fig. Absolute flows (gWWm⁻²month⁻¹) produced by the different functional group at trophic level 3. Mean \pm SE values per mesocosm are given (n = 3). Significant interactions or main effects (p < 0.05) within functional groups are based on two-way ANOVAs (df = 1,8) and are indicated with asterisks. Means with different lowercase letters indicate significant difference among treatments based on posthoc tests corrected for false discovery rate and done separately for different functional groups. No Sig = no significance. See <u>S5 Table</u> for statistical test outcomes.

(TIF)

S3 Fig. The Finn's cycling index expresses the amount of detritus that is recycled relative to the total throughput of the system. Mean \pm SE values per mesocosm are given (n = 3). Significant effects (p < 0.05) are based on two-way ANOVAs with *OA* and *T* (df = 1,8) and are indicated with asterisks. Means with different lower case letters indicate significant difference among treatments. See S3 Table for statistical test outcomes. *OA*, elevated CO₂; *T*, elevated

temperature. (TIF)

S4 Fig. The different structural components of the mesocosm used for this experiment. Each mesocosm comprises 4 'rocky reef' patches (A) and 4 'artificial seagrass' patches (B). The space in between and around these patches was considered 'open sand' habitat (C). The incoming seawater was led into 2 header tanks (800 L) at the beginning of the flow-through facilities, and from there gravity-fed into each mesocosm (D). The header tank was preconditioned to future pCO_2 levels with pure CO_2 (control system ACQ110 Aquatronica, Italy) prior to supplying the water to the 6 acidified mesocosms. In addition, continuous water circulation (approximately 1,800 L per hour) was maintained between each mesocosm and a 60-L supporting bin positioned next to each mesocosm that was bubbled heavily with enriched air at 1,000 ppm pCO₂ (PEGAS 4000 MF Gas Mixer, Columbus Instruments, Columbus, Ohio) or ambient air at 400 ppm pCO_2 , depending on the acidification treatment. These bins also contained the submersible titanium heaters for the T treatments. A diffuser pipe was used to generate a mild circular current inside the mesocosms using the water exchange between supporting bin and mesocosm and alternating direction every 6 hours (E). A filter column (approximately 20 µm) allowed water to flow back into the 60-L bin by gravity (F) and ensured that larger organisms were always retained within the mesocosms. In summary, this technically complex set-up ensured a mesocosm environment without unnatural disturbances such as pump noise, air bubbles, or electrical currents. A 250W metal halide lamp (Osram Powerstar HQI-T 250/D/PRO) mounted just above the mesocosm (G) ensured an irradiance that corresponded to approximately 6–7 m water depth in Gulf St. Vincent (Phillips et al. 1981). T, elevated temperature.

(TIF)

S5 Fig. Variability in pH and temperature over the 5-month study period. This includes 3 phases: 1) the first week of the acclimation period, 2) the progressive elevation to treatment levels, and 3) at maintained treatment levels. Mean ± SD are shown based on 3 mesocosms for each treatment. pH and temperature were both measured once daily in each mesocosm around midday.

(TIF)

S6 Fig. Diurnal variability in pH measured over a 5-day period in the middle of the study period. This analysis was only done for 1 mesocosm per treatment combination, serving as an example. For these 4 mesocosms in parallel, pH was recorded at 30-minute intervals with an automated pH logger (control system ACQ110 Aquatronica, Italy). (TIF)

S7 Fig. Schematic diagram showing the different phases of model building and execution. A) data collection from the mesocosms and parameter estimation, B) mass-balance modelling in Ecopath, and C) model balancing and validation. (TIF)

S8 Fig. PREBAL of the control and acidification models plotting (a) biomass estimates (gWWm⁻²), (b) production/biomass ratio (per 4 months), and (c) consumption/biomass (per 4 months) on a log scale versus functional groups ranked by their trophic level, from lowest to highest trophic level. A constant of 1 was added to all response variables to avoid some negative values ($Log_{10} [x + 1]$) prior to PREBAL plotting. For specific functional group name, refer to the legend. Herb. = herbivorous. PREBAL is shown only for base models that are built on the average of all the input parameters (B, P/B, Q/B) across mesocosms within

each climate treatment. B, biomass; P/B, production per unit of biomass; Q/B, consumption per unit of biomass.

(TIF)

S9 Fig. PREBAL of the temperature and temperature + acidification models plotting (a) biomass estimates (gWWm⁻²), (b) production/biomass ratio (per 4 months), and (c) consumption/biomass (per 4 months) on a log scale versus functional groups ranked by their trophic level, from lowest to highest trophic level. A constant of 1 was added to all response variables to avoid some negative values ($Log_{10} [x + 1]$) prior to PREBAL plotting. For specific functional group name, refer to the legend. Herb. = herbivorous. PREBAL is shown only for base models that are built on the average of all the input parameters (B, P/B, Q/B) across meso-cosms within each climate treatment. B, biomass; P/B, production per unit of biomass; Q/B, consumption per unit of biomass.

(TIF)

S1 Table. Analysis of variance of the effects of *OA* and *T* and their interaction on absolute flows and transfer efficiency between successive trophic levels of the food web. Significant differences are indicated with asterisks, with *: p < 0.05, **: p < 0.01, and ***: p < 0.001. *OA*, elevated CO₂; *T*, elevated temperature. (XLSX)

S2 Table. Analysis of variance of the effects of OA and *T* and their interaction on living biomass by trophic levels of the food web. Significant differences are indicated with asterisks, with *: p < 0.05, **: p < 0.01, and ***: p < 0.001. *OA*, elevated CO₂; *T*, elevated temperature. (XLSX)

S3 Table. Analysis of variance of the effects of *OA* and *T* and their interaction on cyanobacteria (% cover), flow (%) to detritus, and Finn's cycling index in the food web. Significant differences are indicated with asterisks, with *: p < 0.05, **: p < 0.01, and ***: p < 0.001. *OA*, elevated CO₂; *T*, elevated temperature. (XLSX)

S4 Table. Analysis of variance of the effects of *OA* and *T* and their interaction on the absolute flows of contributing functional groups from trophic level 1 to 2. Functional groups were ordered in terms of their contribution to total energy flows. Significant differences indicated with asterisks, with *: p < 0.05, **: p < 0.01, and ***: p < 0.001. *OA*, elevated CO₂; *T*, elevated temperature.

(XLSX)

S5 Table. Analysis of variance of the effects of *OA* and *T* and their interaction on the absolute flows of contributing functional groups from trophic levels 2 to 3. Functional groups were ordered in terms of their contribution to total energy flows. Significant differences are indicated with asterisks, with *: p < 0.05, **: p < 0.01, and ***: p < 0.001. *OA*, elevated CO₂; *T*, elevated temperature.

(XLSX)

S6 Table. List of species/taxa and their respective functional group considered in the mesocosm food webs.

(XLSX)

S7 Table. Mean (\pm SD) seawater parameters in the experimental mesocosoms with 2 crossed factors of *T* and *OA*. Standard deviations represent the variability between individual

mesocosms. *OA*, elevated CO₂; *T*, elevated temperature. (XLSX)

S8 Table. Input (nonitalic) and output (italic) parameters for the ecosystem components used in C models. TL: trophic level, B: biomass (gWWm⁻²), P/B: production/biomass ratio (per 4 months), Q/B: consumption/biomass ratio (per 4 months), EE: ecotrophic efficiency, P/Q: gross food conversion efficiency of each functional group. CM1 represents control model 1, CM2 represents control model 2, and CM3 represents control model 3. The value in bold indicates that particular parameter varied among the models and mentioned below the table. C, control.

(XLSX)

S9 Table. Input (nonitalic) and output (italic) parameters for the ecosystem components used in the *OA* **models.** TL: trophic level, B: biomass (gWWm⁻²), P/B: production/biomass ratio (per 4 months), Q/B: consumption/biomass ratio (per 4 months), EE: ecotrophic efficiency, P/Q: gross food conversion efficiency of each functional group. OAM1 represents acid-ification model 1, OAM2 represents acidification model 2, and OAM3 represents acidification model 3. The value in bold indicates that particular parameter varies among the models as mentioned below the table. *OA*, elevated CO₂. (XLSX)

S10 Table. Input (nonitalic) and output (italic) parameters for the ecosystem components used in the *T* **models. TL: trophic level, B: biomass (gWWm⁻²), P/B: production/biomass ratio (per 4 months), Q/B: consumption/biomass ratio (per 4 months), EE: ecotrophic efficiency, P/Q: gross food conversion efficiency of each functional group. TM1 represents temperature model 1, TM2 represents temperature model 2, and TM3 represents temperature model 3. The values in bold indicate that particular parameters vary among the models as mentioned below the table.** *T***, elevated temperature. (XLSX)**

S11 Table. Input (nonitalic) and output (italic) parameters for the ecosystem components used in the OAT models. TL: trophic level, B: biomass (gWWm⁻²), P/B: production/biomass ratio (per 4 months), Q/B: consumption/biomass ratio (per 4 months), EE: ecotrophic efficiency, P/Q: gross food conversion efficiency of each functional group. OATM1 represents temperature and acidification model 1, OATM2 represents temperature and acidification model 2, and OATM3 represents temperature and acidification model 3. The value in bold indicates that particular parameter varies among the models as mentioned below the table. *OAT*, elevated CO₂ and temperature. (XLSX)

S12 Table. Predator/prey matrix (column/row) for C models. The fraction of one compartment consumed by another is expressed as the fraction of the total diet, the sum of each column being equal to 1. Values with mean ± SD represent the adjustment of different prey groups in predator diets across models. C, control. (XLSX)

S13 Table. Predator/prey matrix (column/raw) for *OA* **models.** The fraction of one compartment consumed by another is expressed as the fraction of the total diet, the sum of each column being equal to 1. Values with mean \pm SD represent the adjustment of different prey groups in predator diets across models. *OA*, elevated CO₂. (XLSX)

S14 Table. Predator/prey matrix (column/raw) for *T* **models.** The fraction of one compartment consumed by another is expressed as the fraction of the total diet, the sum of each column being equal to 1. Values with mean \pm SD represent the adjustment of different prey groups in predator diets across models. *T*, elevated temperature. (XLSX)

S15 Table. Predator/prey matrix (column/raw) for OAT models. The fraction of one compartment consumed by another is expressed as the fraction of the total diet, the sum of each column being equal to 1. Values with mean \pm SD represent the adjustment of different prey groups in predator diets across models. OAT, elevated CO₂ and temperature. (XLSX)

S16 Table. Source of additional information used ($\sqrt{}$) to parameterize base (control model) models for different functional groups, where similar values used across treatments were specified in the supplementary text.

(XLSX)

S1 Text. Supplementary methods. (DOCX)

S1 Data. Data used to generate the manuscript figures. (XLSX)

Acknowledgments

We would like to thank W. Hutchinson and M. Gluis for their technical advice and supports during the mesocosm experiment. We acknowledge the effort from all the students who helped with the construction and maintenance of the mesocosm, with special thanks to Camilo Ferreira and Georgia Walden, Emma Marangon, and Angélique Bonnet. We also thank T. Kildea for providing local temperature data.

Author Contributions

Conceptualization: Hadayet Ullah, Ivan Nagelkerken, Silvan U. Goldenberg. Data curation: Hadayet Ullah, Silvan U. Goldenberg. Formal analysis: Hadayet Ullah. Funding acquisition: Ivan Nagelkerken. Investigation: Hadayet Ullah, Silvan U. Goldenberg. Methodology: Hadayet Ullah. Resources: Ivan Nagelkerken. Supervision: Ivan Nagelkerken, Damien A. Fordham. Validation: Hadayet Ullah. Writing – original draft: Hadayet Ullah, Ivan Nagelkerken, Damien A. Fordham. Writing – review & editing: Hadayet Ullah, Ivan Nagelkerken, Silvan U. Goldenberg, Damien A. Fordham.

References

- Mouquet N, Lagadeuc Y, Devictor V, Doyen L, Duputié A, et al. (2015) REVIEW: Predictive ecology in a changing world. Journal of Applied Ecology 52: 1293–1310.
- 2. Nagelkerken I, Connell SD (2015) Global alteration of ocean ecosystem functioning due to increasing human CO2 emissions. Proceedings of the National Academy of Sciences 112: 13272–13277.
- Nagelkerken I, Goldenberg SU, Ferreira CM, Russell BD, Connell SD (2017) Species Interactions Drive Fish Biodiversity Loss in a High-CO2 World. Current Biology 27: 2177–2184.e2174. <u>https://doi.org/10.1016/j.cub.2017.06.023</u> PMID: 28690109
- O'Connor MI, Piehler MF, Leech DM, Anton A, Bruno JF (2009) Warming and Resource Availability Shift Food Web Structure and Metabolism. PLoS Biol 7: e1000178. <u>https://doi.org/10.1371/journal.pbio.1000178</u> PMID: <u>19707271</u>
- 5. Hoegh-Guldberg O, Bruno JF (2010) The Impact of Climate Change on the World's Marine Ecosystems. Science 328: 1523–1528. <u>https://doi.org/10.1126/science.1189930</u> PMID: 20558709
- Jackson JBC, Kirby MX, Berger WH, Bjorndal KA, Botsford LW, et al. (2001) Historical Overfishing and the Recent Collapse of Coastal Ecosystems. Science 293: 629–637. <u>https://doi.org/10.1126/science. 1059199</u> PMID: <u>11474098</u>
- Doney SC, Ruckelshaus M, Duffy JE, Barry JP, Chan F, et al. (2012) Climate Change Impacts on Marine Ecosystems. Annual Review of Marine Science 4: 11–37. <u>https://doi.org/10.1146/annurevmarine-041911-111611</u> PMID: 22457967
- Lurgi M, López BC, Montoya JM (2012) Novel communities from climate change. Philosophical Transactions of the Royal Society B: Biological Sciences 367: 2913–2922.
- Finn JT (1976) Measures of ecosystem structure and function derived from analysis of flows. Journal of Theoretical Biology 56: 363–380. PMID: <u>944838</u>
- Tylianakis JM, Tscharntke T, Lewis OT (2007) Habitat modification alters the structure of tropical hostparasitoid food webs. Nature 445: 202–205. <u>https://doi.org/10.1038/nature05429</u> PMID: <u>17215842</u>
- Wernberg T, Bennett S, Babcock RC, de Bettignies T, Cure K, et al. (2016) Climate-driven regime shift of a temperate marine ecosystem. Science 353: 169–172. <u>https://doi.org/10.1126/science.aad8745</u> PMID: <u>27387951</u>
- 12. Brook BW, Sodhi NS, Bradshaw CJA (2008) Synergies among extinction drivers under global change. Trends in Ecology & Evolution 23: 453–460.
- Yvon-Durocher G, Allen AP, Cellamare M, Dossena M, Gaston KJ, et al. (2015) Five Years of Experimental Warming Increases the Biodiversity and Productivity of Phytoplankton. PLoS Biol 13: e1002324. https://doi.org/10.1371/journal.pbio.1002324 PMID: 26680314
- 14. Poloczanska ES, Burrows MT, Brown CJ, García Molinos J, Halpern BS, et al. (2016) Responses of Marine Organisms to Climate Change across Oceans. Frontiers in Marine Science 3.
- Pörtner HO, Knust R (2007) Climate Change Affects Marine Fishes Through the Oxygen Limitation of Thermal Tolerance. Science 315: 95–97. <u>https://doi.org/10.1126/science.1135471</u> PMID: <u>17204649</u>
- 16. Dillon ME, Wang G, Huey RB (2010) Global metabolic impacts of recent climate warming. Nature 467: 704–706. https://doi.org/10.1038/nature09407 PMID: 20930843
- Pistevos JCA, Nagelkerken I, Rossi T, Olmos M, Connell SD (2015) Ocean acidification and global warming impair shark hunting behaviour and growth. 5: 16293.
- Pörtner H (2008) Ecosystem effects of ocean acidification in times of ocean warming: a physiologist's view. Marine Ecology Progress Series 373: 203–217.
- Kroeker KJ, Kordas RL, Crim R, Hendriks IE, Ramajo L, et al. (2013) Impacts of ocean acidification on marine organisms: quantifying sensitivities and interaction with warming. Global Change Biology 19: 1884–1896. <u>https://doi.org/10.1111/gcb.12179</u> PMID: 23505245
- Nagelkerken I, Munday PL (2016) Animal behaviour shapes the ecological effects of ocean acidification and warming: moving from individual to community-level responses. Global Change Biology 22: 974– 989. <u>https://doi.org/10.1111/gcb.13167</u> PMID: <u>26700211</u>
- Walther G-R (2010) Community and ecosystem responses to recent climate change. Philosophical Transactions of the Royal Society B: Biological Sciences 365: 2019–2024. <u>https://doi.org/10.1098/rstb.</u> 2010.0021 PMID: 20513710
- 22. Van der Putten WH, Macel M, Visser ME (2010) Predicting species distribution and abundance responses to climate change: why it is essential to include biotic interactions across trophic levels. Philosophical Transactions of the Royal Society B: Biological Sciences 365: 2025–2034.
- 23. Vergés A, Steinberg PD, Hay ME, Poore AGB, Campbell AH, et al. (2014) The tropicalization of temperate marine ecosystems: climate-mediated changes in herbivory and community phase shifts. Proceedings of the Royal Society B: Biological Sciences 281.

- 24. Woodward G, Benstead JP, Beveridge OS, Blanchard J, Brey T, et al. (2010) Ecological Networks in a Changing Climate. Advances in Ecological Research 42: 71–138.
- Sommer U, Lengfellner K (2008) Climate change and the timing, magnitude, and composition of the phytoplankton spring bloom. Global Change Biology 14: 1199–1208.
- **26.** Brown CJ, Fulton EA, Hobday AJ, Matear RJ, Possingham HP, et al. (2010) Effects of climate-driven primary production change on marine food webs: implications for fisheries and conservation. Global Change Biology 16: 1194–1212.
- Connell SD, Russell BD (2010) The direct effects of increasing CO2 and temperature on non-calcifying organisms: increasing the potential for phase shifts in kelp forests. Proceedings of the Royal Society B: Biological Sciences 277: 1409–1415. <u>https://doi.org/10.1098/rspb.2009.2069</u> PMID: 20053651
- 28. Goldenberg SU, Nagelkerken I, Ferreira CM, Ullah H, Connell SD Boosted food web productivity through ocean acidification collapses under warming. Global Change Biology: n/a–n/a.
- Hansson L-A, Nicolle A, Graneli W, Hallgren P, Kritzberg E, et al. (2013) Food-chain length alters community responses to global change in aquatic systems. Nature Clim Change 3: 228–233.
- Diaz RJ, Rosenberg R (2008) Spreading Dead Zones and Consequences for Marine Ecosystems. Science 321: 926–929. <u>https://doi.org/10.1126/science.1156401</u> PMID: <u>18703733</u>
- Davis JM, Rosemond AD, Eggert SL, Cross WF, Wallace JB (2010) Long-term nutrient enrichment decouples predator and prey production. Proceedings of the National Academy of Sciences 107: 121– 126.
- Brown JH, Gillooly JF, Allen AP, Savage VM, West GB (2004) TOWARD A METABOLIC THEORY OF ECOLOGY. Ecology 85: 1771–1789.
- Brose U, Williams RJ, Martinez ND (2006) Allometric scaling enhances stability in complex food webs. Ecology Letters 9: 1228–1236. <u>https://doi.org/10.1111/j.1461-0248.2006.00978.x</u> PMID: <u>17040325</u>
- Sommer U, Paul C, Moustaka-Gouni M (2015) Warming and Ocean Acidification Effects on Phytoplankton-From Species Shifts to Size Shifts within Species in a Mesocosm Experiment. PLoS ONE 10: 17.
- Loreau M (2010) From Populations to Ecosystems. Theoretical Foundations for a New Ecological Synthesis (MPB-46): Princeton University Press.
- Fordham DA (2015) Mesocosms Reveal Ecological Surprises from Climate Change. PLoS Biol 13: e1002323. https://doi.org/10.1371/journal.pbio.1002323 PMID: 26680131
- Stocker T (2014) Climate change 2013: the physical science basis: Working Group I contribution to the Fifth assessment report of the Intergovernmental Panel on Climate Change: Cambridge University Press.
- Pauly D, Christensen V, Walters C (2000) Ecopath, Ecosim, and Ecospace as tools for evaluating ecosystem impact of fisheries. ICES Journal of Marine Science 57: 697–706.
- **39.** Thompson RM, Brose U, Dunne JA, Hall RO Jr., Hladyz S, et al. Food webs: reconciling the structure and function of biodiversity. Trends in Ecology & Evolution 27: 689–697.
- O'Connor MI (2009) Warming strengthens an herbivore–plant interaction. Ecology 90: 388–398. PMID: 19323223
- Carr LA, Bruno JF (2013) Warming increases the top-down effects and metabolism of a subtidal herbivore. PeerJ 1: e109. <u>https://doi.org/10.7717/peerj.109</u> PMID: 23904988
- Gilman SE, Urban MC, Tewksbury J, Gilchrist GW, Holt RD (2010) A framework for community interactions under climate change. Trends in Ecology & Evolution 25: 325–331.
- Alexander JM, Diez JM, Hart SP, Levine JM When Climate Reshuffles Competitors: A Call for Experimental Macroecology. Trends Ecol Evol 31: 831–841. <u>https://doi.org/10.1016/j.tree.2016.08.003</u> PMID: <u>27640784</u>
- 44. Lawton JH (1996) The Ecotron Facility at Silwood Park: The Value of "Big Bottle" Experiments. Ecology 77: 665–669.
- 45. Mertens NL, Russell BD, Connell SD (2015) Escaping herbivory: ocean warming as a refuge for primary producers where consumer metabolism and consumption cannot pursue. Oecologia 179: 1223–1229. https://doi.org/10.1007/s00442-015-3438-8 PMID: 26363905
- 46. Barott KL, Rodriguez-Brito B, Janouškovec J, Marhaver KL, Smith JE, et al. (2011) Microbial diversity associated with four functional groups of benthic reef algae and the reef-building coral Montastraea annularis. Environmental Microbiology 13: 1192–1204. <u>https://doi.org/10.1111/j.1462-2920.2010.</u> 02419.x PMID: 21272183
- Charpy L, Palinska KA, Abed RMM, Langlade MJ, Golubic S (2012) Factors influencing microbial mat composition, distribution and dinitrogen fixation in three western Indian Ocean coral reefs. European Journal of Phycology 47: 51–66.

- **48.** O'Neil JM, Davis TW, Burford MA, Gobler CJ (2012) The rise of harmful cyanobacteria blooms: The potential roles of eutrophication and climate change. Harmful Algae 14: 313–334.
- Bender D, Diaz-Pulido G, Dove S (2014) Warming and acidification promote cyanobacterial dominance in turf algal assemblages. Marine Ecology Progress Series 517: 271–284.
- D'Angelo C, Wiedenmann J (2014) Impacts of nutrient enrichment on coral reefs: new perspectives and implications for coastal management and reef survival. Current Opinion in Environmental Sustainability 7: 82–93.
- HW ÂÂP, Joyner JJ, Joyner AR, Arthur K, Paul V, et al. (2008) Co-occurrence of dinoflagellate and cyanobacterial harmful algal blooms in southwest Florida coastal waters: dual nutrient (N and P) input controls. Marine Ecology Progress Series 371: 143–153.
- Paerl HW, Paul VJ (2012) Climate change: Links to global expansion of harmful cyanobacteria. Water Research 46: 1349–1363. <u>https://doi.org/10.1016/j.watres.2011.08.002</u> PMID: <u>21893330</u>
- Nagle DG, Paul VJ (1998) Chemical defense of a marine cyanobacterial bloom. Journal of Experimental Marine Biology and Ecology 225: 29–38.
- López-Urrutia Á, San Martin E, Harris RP, Irigoien X (2006) Scaling the metabolic balance of the oceans. Proceedings of the National Academy of Sciences 103: 8739–8744.
- 55. Yvon-Durocher G, Jones JI, Trimmer M, Woodward G, Montoya JM (2010) Warming alters the metabolic balance of ecosystems. Philosophical Transactions of the Royal Society of London B: Biological Sciences 365: 2117–2126. <u>https://doi.org/10.1098/rstb.2010.0038</u> PMID: 20513719
- Johansen JL, Pratchett MS, Messmer V, Coker DJ, Tobin AJ, et al. (2015) Large predatory coral trout species unlikely to meet increasing energetic demands in a warming ocean. 5: 13830.
- 57. Allen AP, Gillooly JF (2007) The mechanistic basis of the metabolic theory of ecology. Oikos 116: 1073–1077.
- O'Connor MP, Kemp SJ, Agosta SJ, Hansen F, Sieg AE, et al. (2007) Reconsidering the mechanistic basis of the metabolic theory of ecology. Oikos 116: 1058–1072.
- 59. Odum WE, Heald EJ (1975) The detritus-based food web of an. Estuar Res Chem Biol Estuar Syst 1: 265.
- Vasconcellos M, Mackinson S, Sloman K, Pauly D (1997) The stability of trophic mass-balance models of marine ecosystems: a comparative analysis. Ecological Modelling 100: 125–134.
- 61. Odum EP (1969) The Strategy of Ecosystem Development. Science 164: 262–270. PMID: 5776636
- Nagelkerken I, Russell BD, Gillanders BM, Connell SD (2016) Ocean acidification alters fish populations indirectly through habitat modification. Nature Clim Change 6: 89–93.
- Connell SD, Doubleday ZA, Hamlyn SB, Foster NR, Harley CDG, et al. (2017) How ocean acidification can benefit calcifiers. Current Biology 27: R95–R96. <u>https://doi.org/10.1016/j.cub.2016.12.004</u> PMID: 28171763
- 64. Woodward G, Brown LE, Edwards FK, Hudson LN, Milner AM, et al. (2012) Climate change impacts in multispecies systems: drought alters food web size structure in a field experiment. Philosophical Transactions of the Royal Society of London B: Biological Sciences 367: 2990–2997. <u>https://doi.org/10.1098/ rstb.2012.0245</u> PMID: 23007087
- Hairston NG (1993) Cause-Effect Relationships in Energy Flow, Trophic Structure, and Interspecific Interactions. The American Naturalist 142: 379–411.
- Dickman EM, Newell JM, González MJ, Vanni MJ (2008) Light, nutrients, and food-chain length constrain planktonic energy transfer efficiency across multiple trophic levels. Proceedings of the National Academy of Sciences 105: 18408–18412.
- Pauly D, Christensen V, Dalsgaard J, Froese R, Torres F (1998) Fishing Down Marine Food Webs. Science 279: 860–863. PMID: 9452385
- 68. Christensen V, Pauly D (1993) Trophic models of aquatic ecosystems: WorldFish.
- Christensen V, Pauly D (1992) ECOPATH II—a software for balancing steady-state ecosystem models and calculating network characteristics. Ecological Modelling 61: 169–185.
- 70. Wiedner C, Rücker J, Brüggemann R, Nixdorf B (2007) Climate change affects timing and size of populations of an invasive cyanobacterium in temperate regions. Oecologia 152: 473–484. <u>https://doi.org/10.1007/s00442-007-0683-5</u> PMID: <u>17375336</u>
- **71.** Bif MB, Yunes JS (2017) Distribution of the marine cyanobacteria Trichodesmium and their association with iron-rich particles in the South Atlantic Ocean. Aquatic Microbial Ecology 78: 107–119.
- Glibert PM, Heil CA, Hollander D, Revilla M, Hoare A, et al. (2004) Evidence for dissolved organic nitrogen and phosphorus uptake during a cyanobacterial bloom in Florida Bay. Marine Ecology Progress Series 280: 73–83.

- **73.** Albert S, O'Neil JM, Udy JW, Ahern KS, O'Sullivan CM, et al. (2005) Blooms of the cyanobacterium Lyngbya majuscula in coastal Queensland, Australia: disparate sites, common factors. Marine Pollution Bulletin 51: 428–437. <u>https://doi.org/10.1016/j.marpolbul.2004.10.016</u> PMID: <u>15757741</u>
- Bryars S, Rowling K (2009) Benthic Habitats of Eastern Gulf St Vincent: Major Changes in Benthic Cover and Composition Following European Settlement of Adelaide. Transactions of the Royal Society of South Australia 133: 318–338.
- 75. Makoto O, Tsutomu I (1984) Methods in marine zooplankton ecology.
- **76.** Jeffrey SW, Humphrey GF (1975) New spectrophotometric equations for determining chlorophylls a, b, c1 and c2 in higher plants, algae and natural phytoplankton. Biochemie und Physiologie der Pflanzen 167: 191–194.
- Kohler KE, Gill SM (2006) Coral Point Count with Excel extensions (CPCe): A Visual Basic program for the determination of coral and substrate coverage using random point count methodology. Computers & Geosciences 32: 1259–1269.
- Christensen V, Walters CJ, Pauly D (2005) Ecopath with Ecosim: a user's guide. Fisheries Centre, University of British Columbia, Vancouver 154.
- 79. Lindeman RL (1942) The Trophic-Dynamic Aspect of Ecology. Ecology 23: 399–417.
- 80. Ulanowicz RE (1997) Ecology, the Ascendent Perspective: Columbia University Press.
- 81. Ulanowicz RE (2012) Growth and development: ecosystems phenomenology: Springer Science & Business Media.
- Heymans JJ, Coll M, Libralato S, Christensen V (2011) 9.06—Ecopath Theory, Modeling, and Application to Coastal Ecosystems. Treatise on Estuarine and Coastal Science. Waltham: Academic Press. pp. 93–113.
- 83. Morissette L (2007) Complexity, cost and quality of ecosystem models and their impact on resilience: a comparative analysis, with emphasis on marine mammals and the Gulf of St. Lawrence: University of British Columbia.
- Benjamini Y, Hochberg Y (1995) Controlling the False Discovery Rate: A Practical and Powerful Approach to Multiple Testing. Journal of the Royal Statistical Society Series B (Methodological) 57: 289–300.
- 85. Team RC (2015) R: a language and environment for statistical computing. Vienna, Austria; 2015.